Author’s response to reviews

Title: A cross-sectional survey of parental attitudes towards Human papillomavirus vaccination exclusion categories in Brazil

Authors:

Mariana Gattegno (mvgattegno@gmail.com)

Maria Vertamatti (mdorafig@gmail.com)

Robert Bednarczyk (robert.bednarczyk@emory.edu)

Dabney Evans (dabney.evans@emory.edu)

Version: 2 Date: 04 Feb 2019

Author’s response to reviews:

-- We note that the order of authors in the system is not the same as in the title page of the manuscript. Please ensure that the order of authors in the system is the same as in the title page of the manuscript. If you need to change your author list from the order currently on the title page of your manuscript you will need to fill out a change in authorship form and send it to the Editorial office to be approved by the Editor. Please see attached.

There was no attachment to your email.

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Text to this effect has been added.

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Text to this effect has been added.

-- As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

Text noting this affiliation has been added.
Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

This document has been removed.

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Revised manuscript has been uploaded.